Author's response to reviews

Title: Discharge instructions for caregivers in the context of pediatric emergency care: A Narrative Synthesis.

Authors:

Janet A Curran (Janet.Curran@iwk.nshealth.ca)
Andrea Murphy (Andrea.murphy@dal.ca)
Mandi Newton (Mandi.newton@ualberta.ca)
Roger Zemek (rzemek@cheo.on.ca)
Lisa Hartling (hartling@ualberta.ca)
Amy Plint (plint@cheo.on.ca)
Jill Chorney (Jill.chorney@iwk.nshealth.ca)
Shannon MacPhee (Shannon.macphee@iwk.nshealth.ca)
Samuel Campbell (Samuel.campbell@cdha.nshealth.ca)
Mona Jabbour (jabbour@che.on.ca)
David Petrie (Davida.petrie@cdha.nshealth.ca)
Randy Colwell (jrcolwell@gmail.com)
Kate MacWilliams (mmacwill@dal.ca)
Alicia Nolan (Alicia.Nolan@iwk.nshealth.ca)

Version: 5 Date: 12 February 2014

Author's response to reviews: see over
February 1, 2014

Dear Editor,


Thank you for the opportunity to strengthen our manuscript with your critical feedback. Please find below our response to your comments.

Editor’s comments
1. We have included our Prospero Registration number at the end of the Abstract.
2. Each authors individual contribution has been outlined in Authors Contribution.
3. Figure title and legend section have been added after the Reference list.

Reviewer’s comments
1. We have changed scoping review to scoping search.
2. We will include observation studies. All studies examining interventions or processes related to discharge instructions in pediatric emergency departments will be included (experimental, quasi-experimental, observational and qualitative)
3. Rational has been replaced with rationale
4. References 1, 24, and 26 have been revised
5. Additional description of quality appraisal strategy has been added to Methods section.

Thank you again for taking the time to review our manuscript and we trust we have adequately addressed your concerns.

Regards
Janet Curran